



# **State of the Art: Lung Cancer Staging Using Updated Imaging Modalities**

Nihal M. Batouty <sup>1,†</sup><sup>®</sup>, Gehad A. Saleh <sup>1,†</sup><sup>®</sup>, Ahmed Sharafeldeen <sup>2</sup><sup>®</sup>, Heba Kandil <sup>2,3</sup><sup>®</sup>, Ali Mahmoud <sup>2</sup><sup>®</sup>, Ahmed Shalaby <sup>2</sup><sup>®</sup>, Maha Yaghi <sup>4</sup><sup>®</sup>, Adel Khelifi <sup>5</sup><sup>®</sup>, Mohammed Ghazal <sup>4</sup><sup>®</sup> and Ayman El-Baz <sup>2,\*</sup><sup>®</sup>

- <sup>1</sup> Department of Diagnostic and Interventional Radiology, Faculty of Medicine, Mansoura University, Mansoura 35516, Egypt
- <sup>2</sup> Bioengineering Department, University of Louisville, Louisville, KY 40292, USA
- <sup>3</sup> Information Technology Department, Faculty of Computers and Informatics, Mansoura University, Mansoura 35516, Egypt
- <sup>4</sup> Electrical, Computer, and Biomedical Engineering Department, Abu Dhabi University, Abu Dhabi 59911, United Arab Emirates
- <sup>5</sup> Computer Science and Information Technology Department, Abu Dhabi University, Abu Dhabi 59911, United Arab Emirates
- \* Correspondence: aselba01@louisville.edu
- + These authors contributed equally to this work.

Abstract: Lung cancer is among the most common mortality causes worldwide. This scientific article is a comprehensive review of current knowledge regarding screening, subtyping, imaging, staging, and management of treatment response for lung cancer. The traditional imaging modality for screening and initial lung cancer diagnosis is computed tomography (CT). Recently, a dual-energy CT was proven to enhance the categorization of variable pulmonary lesions. The National Comprehensive Cancer Network (NCCN) recommends usage of fluorodeoxyglucose positron emission tomography (FDG PET) in concert with CT to properly stage lung cancer and to prevent fruitless thoracotomies. Diffusion MR is an alternative to FDG PET/CT that is radiation-free and has a comparable diagnostic performance. For response evaluation after treatment, FDG PET/CT is a potent modality which predicts survival better than CT. Updated knowledge of lung cancer genomic abnormalities and treatment regimens helps to improve the radiologists' skills. Incorporating the radiologic experience is crucial for precise diagnosis, therapy planning, and surveillance of lung cancer.

Keywords: chest; lung cancer; diffusion-weighted MRI; computed tomography (CT); FDG PET/CT

## 1. Introduction

Lung cancer is among the most frequent mortality causes worldwide [1]. Patients suffering from lung cancer are first diagnosed by detecting a solitary pulmonary nodule (SPN) or a pulmonary mass by computed tomography (CT) or chest radiography at an early stage of the disease. Pulmonary nodules are widespread medical challenges ranging from metastasis and lung cancer to infections and other benign lesions for which management can be complex [2]. SPNs have a 30–40% chance of being malignant [3]. Precise categorization of SPNs is clinically crucial because the early diagnosis of malignant disease would increase the chances of a successful surgical resection and a 5-year survival rate [4].

The Lung Imaging Reporting and Data System (Lung-RADS) was first published by the American College of Radiology (ACR) in 2014 to enable standardized reporting and management of abnormal findings in lung cancer screening [5,6].

The ACR updated the Lung-RADS version 1.1 in May 2019, with the two main changes concerning perifissural nodules and ground-glass nodules as a negative screening result [7,8]. Lung-RADS version 1.1 is better at categorizing the solid pulmonary nodules and reduces the percentage of false positives by downgrading perifissural nodules in



Citation: Batouty, N.M.; Saleh, G.A.; Sharafeldeen, A.; Kandil; H.; Mahmoud, A.; Shalaby, A.; Yaghi, M.; Khelifi, A.; Ghazal, M.; El-Baz, A. State of the Art: Lung Cancer Staging Using Updated Imaging Modalities. *Bioengineering* **2022**, *9*, 493. https://doi.org/10.3390/ bioengineering9100493

Academic Editor: Zhen Cheng

Received: 7 September 2022 Accepted: 16 September 2022 Published: 22 September 2022

**Publisher's Note:** MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



**Copyright:** © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the range of 6–10 mm to benign lesions [8]. The Fleischner Society recommendations for the management of incidental SPNs were initially released in 2005, then further distinct guidelines for subsolid nodules were announced in 2013 [9,10].

In 2017, the latest guidelines for solid and subsolid nodules were collected in a simple table, while special recommendations were incorporated for multiple nodules. The minimum threshold size for solid nodules on routine follow-up examination was increased. Additionally, fewer follow-up assessments were proposed for stable nodules. Furthermore, a longer period was advised in the case of subsolid nodules before the primary follow-up. In addition, the follow-up period was increased to 5 years. The previous recommendations do not apply to immunosuppressed patients, lung cancer screening, or primary cancer patients [11].

This comprehensive review aims to discuss the current knowledge regarding screening, subtyping, imaging, staging, and management of treatment response for lung cancer. The updated knowledge of lung cancer genomic abnormalities and treatment protocols is now considered crucial for radiologists for precise diagnosis and as a part of a multidisciplinary team for proper management and therapy planning for lung cancer patients.

The rest of this paper is organized as follows: Section 2 discusses the classification of lung cancer. Section 3 demonstrates lung cancer screening. Section 4 illustrates TNM lung cancer staging. Section 5 presents lung cancer follow-up and response evaluation. Section 6 shows World Health Organization (WHO) criteria and Response Evaluation Criteria in Solid Tumors (RECIST). Section 7 demonstrates advances in lung cancer tumor genomics and precision therapy. Finally, Section 8 summarizes the paper.

## 2. Classification of Lung Cancer

## 2.1. Anatomic Classification

Lung cancer has two anatomic types, central and peripheral [12]. Central lung cancer commonly involves squamous cell carcinoma and small-cell carcinoma. Peripheral lung cancer commonly involves adenocarcinoma as well as large-cell carcinoma.

## 2.2. Histopathologic Classification

Following the 2021 World Health Organization (WHO) classification of thoracic tumors (which is similar to the 2015 WHO classification) [13], malignant lung tumors are classified via histopathology into (1) small-cell lung cancer (SCLC), which comprises 15–20% of malignant lung tumors, including small-cell carcinoma, (2) non-small-cell lung cancer (NSCLC), comprising 80–85% of malignant lung tumors, and (3) other, rare types of malignant lung tumors such as sarcoma and lymphoma.

The histopathologic classification of lung cancers helps in deciding treatment and prediction of prognosis [12]. NSCLC represents 85% of lung cancer patients, including squamous cell carcinoma (SCC) (40%), adenocarcinoma (ADC) (50%), and large-cell lung cancer (10%), each of which has a different prognosis [14,15]. Therefore, it is valuable to precisely differentiate between the subtypes of NSCLC before starting treatment [16].

#### 2.2.1. Small-Cell Lung Cancer (SCLC)

The SCLC/small-cell carcinoma is developed from submucosal neuroendocrine cells and is strongly linked to cigarette smoking. This type of cancer is considered the most aggressive type of primary malignant tumor of the lung, with high rate of extrathoracic metastases, which frequently exist at time of the diagnosis [17]. This type of lung cancer is usually linked to massive mediastinal and hilar lymphadenopathy and peripheral lung collapse. Superior vena cava obstruction or infiltration may occur. SCLC is chemotherapysensitive and might be related to paraneoplastic syndrome [12,18,19]. It has a central location in 90% of patients, and appears on X-ray as hilar or para-hilar shadow with wide mediastinum. CT appearance may simulate that of lymphoma due to massive mediastinal lymphadenopathy, but in SCLC, central necrosis of nodes is common [12].

## 2.2.2. Squamous Cell Carcinoma

The squamous cell carcinoma is derived from the bronchial epithelium and is related to cigarette smoking. It shows the slowest growth rate among malignant lung tumors. Squamous cell carcinoma has a central location and is associated with frequent peripheral lobar or segmental collapse. Cavitation with thick irregular wall is common in 82% of patients. It is the most widespread histopathologic category of superior sulcus tumor (Pancoast tumor) [12,19].

## 2.2.3. Adenocarcinoma

The adenocarcinoma type of lung cancer, derived from bronchial submucosal glands, is considered the most widespread histopathologic type of lung cancer in nonsmokers and young women [20]. It appears mostly as a pulmonary nodule and less frequently as a mass (diameter more than 3 cm) with a peripheral location in 72% of the cases (Figure 1). Cavitation is rare with this type of lung cancer. Adenocarcinoma can present in CT by ground-glass opacity surrounding a lung nodule (subsolid nodule), as multiple faint nodules, or, least likely, as consolidation in one lobe or in the whole lung [12,19]. Medi-astinal and hilar nodal spread can occur, but in smaller volume than SCLC. Spread occurs in the same lung or the contralateral lung. Distant metastasis to the brain, bone, liver, and suprarenal glands is frequent [21].



**Figure 1.** A 64-year-old male cigarette smoker with bronchogenic adenocarcinoma. Axial CT image (**A**) reveals spiculated right upper lung lobe mass ( $4.5 \times 2.5$  cm). Axial CT image at distal level (**B**) reveals ipsilateral hilar, mediastinal LN with central necrosis, and pericardial effusion. Axial CT image of upper abdomen (**C**) reveals bilateral metastatic suprarenal masses.

#### 2.2.4. Large-Cell Carcinoma

Large-cell carcinoma is derived from atypical cells that have microscopic appearance of "large" cells, and it is strongly related to cigarette smoking. It has peripheral location in 63% of patients. In CT, it appears as large peripheral masses with early significant mediastinal and hilar nodal spread [12,19].

## 3. Lung Cancer Screening

Screening looks for signs of cancer before the onset of symptoms. This would help in detecting the disease at early stages and hence mitigate adverse events. In the following subsections, different lung cancer screening techniques are presented.

#### 3.1. Digital Chest Radiography

Chest radiography is the most commonly used technique to rule out chest disease and to monitor patients after treatment. Although chest radiography has lower sensitivity of detecting smaller pulmonary lesions than CT, it is easily accessible, has low cost, and uses a low radiation dose [21,22].

Studies of lung cancer screening with conventional chest radiography found that it has a 23% sensitivity rate and a 96% specificity rate [23]. Digital chest radiography has the same specificity with twice or higher increased sensitivity [22].

Even with the advent of digital imaging, the use of chest radiography to detect lung cancer is still considered a real challenge because the diagnostic accuracy of early detection of lung cancer using the digital chest radiography is affected by the confidence and experience level of the reader [22].

Relying only on the chest radiography, 22–63% of malignant lung lesions could be underdiagnosed at a stage of disease that is detectable by CT. Even with the benefit of the advances in digital chest radiography over conventional chest radiography, this large number of false positive results and low detection rates are in support of the low-dosage CT for lung cancer screening [22].

## 3.1.1. Digital Radiography Technique

The acquisition of digital radiography does not differ from that of conventional chest radiography. A direct detector unit is used for obtaining radiographs. Image processing is performed using nonlinear multifrequency band processing. Acquisition parameters differ according to the manufacturer recommendation. Postero-anterior projection only is commonly used in screening programs for lung cancer, even though adding lateral projection leads to better assessment of suspected lesions [22].

#### 3.1.2. Digital Radiography Scale for Lung Cancer Screening

The suspicious focal opacities found in radiographs can be classified into five levels, according to the de Hoop study: negative (level 1), irregular but probably negative (level 2), indeterminate for a lesion (level 3), probable lesion (level 4), and definite lesion (level 5). When radiography findings rated as levels 4–5 are referred for follow-up by CT, malignant lesions are detected in between 37% and 78% of cases (reader-dependent) [22].

#### 3.2. Multi-Detector Computed Tomography (CT)

CT is deemed the routine imaging tool for lung cancer. CT provides the most detailed imaging information about tumor location, size, and extensions [12,24]. Lung cancer in CT appears as circumscribed pulmonary opacity, considered a mass if it measures more than 3 cm in diameter, and a nodule if less than 3 cm [24]. The tumor size is of little diagnostic value to differentiate benign from malignant lung lesions. Commonly, calcifications are criteria of benign lesions, particularly in the case of dense central or laminated calcifications. Yet, malignant lesions may show granular calcifications in 7% of carcinomas, whether they represent tumor calcifications or benign granuloma engulfed by the malignant tumor [19]. Lung cancer usually has an ill-defined, spiculated, irregular margin (Figure 1) and is less likely lobulated with a notch [25]. Unfortunately, ill-defined margins can be seen with benign tumors as well [19].

Lung cancer can present as either peripheral or central tumors. Peripheral tumors are mostly oval or rounded in shape; typical fine strands radiating from tumor margin into the lung, may be seen [17,25]. Associated lung collapse and consolidation is less common than in central tumors, but mucoid impaction or bronchocele may occur distal to tumor obstructing the airways. Air bronchogram and ground-glass opacity are less likely and can occur with broncho-alveolar carcinoma, which may also appear as multiple ill-defined nodules [17,26]. Cavitation with thick irregular wall and fluid level occurs with squamous cell carcinoma [17]. Central tumors are characterized by collapse or consolidation distal to the tumor; this peripheral collapse can show more contrast enhancement than the tumor itself [19,25].

CT assessment of chest wall invasion is unreliable [17]. Obliteration of the extrapleural fat plane or thickened pleura does not always mean chest wall invasion. Rib destruction or extension of pulmonary mass into the chest wall remain the definite signs. Apart from bony invasion, magnetic resonance imaging (MRI) is better than CT in assessment of chest wall

invasion in certain conditions, such as in superior sulcus tumor (Pancoast tumor), an apical lung tumor which invades chest wall and appears as pleural thickening at the lung apex. It can invade the brachial plexus, cervical sympathetic chain, and subclavian vessels. CT can evaluate the relation between the tumor and different mediastinal structures: encasement of great vessels, superior vena cava obstruction, esophagus, and main bronchi [17,19]. Contact with mediastinal structures alone does not indicate mediastinal invasion. Visible tumor tissue within the mediastinal fat is a reliable sign of mediastinal invasion. Contact withmore than 3 cm with mediastinal structures and more than 90% encasement of the aorta are criteria of nonresectable tumors [19,25].

Contrast-enhanced CT provides information about the measured attenuation of lung tumors; it can assess the enhancement of lung tumors which can distinguish benign from malignant lung lesions [27].

CT scans are acquired with patients in the supine position during a single breath hold at the end of inspiration. The following are the imaging parameters; 120 kVp tube voltage, 100–250 mA tube current, and slice thickness and interval varying according to the type of CT scanner (5 or 6 mm in most scanners). Contrast material is injected intravenously (1.5 mL/kg of body weight, rate 3.0 mL/s) via a dual high-pressure injector, then 50 mL saline solution is injected. Images are reconstructed with section thickness 0.6 and slice interval 1 mm [28].

#### 3.3. Dual-Energy Computed Tomography (DECT)

Recently, dual-energy CT (DECT) has been used to enhance characterization of lung lesions. Based on the type of the scanner, there are variable DECT techniques, such as dual-source machines, rapid kVp switch of single source, or dual-layer spectral detectors [29]. Previous studies using phantoms proved the high accuracy of iodine concentration (IC) maps, with slightly higher accurate results for rapid kVp switching and dual-layer spectral detectors [30]. There is high agreement between different vendors for both dual-energy and spectral CT machines, unlike the conventional CT attenuation densities that show less agreement [31]. Thus, IC maps have good potential for clinical routine use [32].

IC maps have several applications that depend on quantitative measurement of iodine density in the region of interest as a marker of vascularity. Accordingly, IC can be considered as a surrogate for tumor tissue angiogenesis and perfusion in different types of dual-energy CT machines. There was good to excellent agreement for both single-source DECT and dual-input CT perfusion parameters including IC when applied to assess the blood supply in patients with lung cancer (intraclass correlation coefficient (ICC) range was 0.82 to 0.99) [33]. IC maps were applied to successfully differentiate the lung cancer; quantitative analysis of iodine uptake images had an added value for the distinction among invasive and minimally invasive or noninvasive adenocarcinoma. Power of diagnosing invasive adenocarcinoma increased from 0.888 to 0.959 (p-value 0.029) after adding iodine uptake parameters [34]. DECT helps to differentiate benign from malignant pulmonary nodules by measurement of iodine value on IC image, with higher sensitivity (92.0% versus 72.0%), similar specificity (70.0% versus 70.0%), and higher diagnostic accuracy (82.2% versus 71.1%) than degree of enhancement measured by conventional post-contrast CT [35]. DECT studies evaluated the contrast enhancement and iodine uptake behavior of pulmonary metastases, which differs based on the primary tumor type [32,36], in addition to evaluating the response of the lung cancer treatment [37].

## 3.3.1. Dual-Energy Computed Tomography (DECT) Technique

Contrast medium is intravenously injected using a 1.2 mL/kg standard dose, a 2–2.5 mL/s flow rate, then 30 mL saline flush at the similar flow rate. Venous phases scanning is obtained after 70 s scan delay time after the beginning of injecting the contrast material. The scanning parameters differ according to the CT scanner type. Scanner vendors provide reconstruction algorithms that could be used to reconstruct conventional and spectral basis images [32].

## 3.3.2. Quantitative Analysis

The spectral images' quantitative analysis is performed using the software provided by the manufacturer. The lesion volume is segmented in the axial plane with soft tissue window in virtual noncontrast images derived from spectral images. Then, it is compared to post-contrast images, resulting in tumor volume reduction. DECT studies recommended this approach to avoid underestimation of the true tumor volume by the exclusion of the peripheral voxels of the tumor to minimize volume averaging [38,39]. The software automatically provides the IC maps and the polychromatic images. IC is measured in mg/mL.

## 3.4. Positron Emission Tomography (PET/CT)

CT is considered the best modality to detect pulmonary lesions; however, it may not be exact enough to distinguish benign and malignant nodules [40,41]. Consequently, invasive procedures (e.g., lung biopsy or video-assisted thoracoscopic surgery) might be performed on patients with benign nodules simply to rule out malignancy [42].

PET/CT using the radiotracer 18F-Fluorodeoxyglucose (18F-FDG) has been verified to be a reliable noninvasive imaging modality to differentiate benign from malignant lesions [43]. The 18F-FDG PET/CT method estimates the rate of cellular glucose metabolism, thus it allows the detection of the metabolically active tissue. The malignant lesions contain metabolically active cells with greater glucose uptake compared with cells in benign tumors [3]. PET/CT combines both metabolic and morphologic information with high sensitivity and specificity in distinguishing malignant pulmonary lesions (96.8% and 77.8%, respectively) [43–45]. Thus, along with its deep-rooted role in staging of pulmonary tumor, PET/CT is a critical diagnostic tool principally used to distinguish between benign and malignant SPNs, decreasing the number of unnecessary invasive procedures for benign SPNs [3].

The National Comprehensive Cancer Network (NCCN) recommends the use of 18F-FDG PET/CT for the proper lung cancer staging and for preventing fruitless thoracotomies [46].

The dispersion of the 18F-FDG uptake in the lesions is not homogeneous. The lesion heterogeneity is related to the cellular proliferation, hypoxia, necrosis, and angiogenesis caused by the malignancy [47]. Therefore, measuring lesion heterogeneity is valuable to differentiate among benign and malignant pulmonary lesions. The tumor heterogeneity of 18F-FDG uptake was proposed to be characterized using PET image texture analysis [48].

## 3.4.1. Patient Preparation

Patients are instructed to fast for 6 hours prior to the PET/CT scan to adjust the level of blood glucose of the fingertip blood sampling to <10 mmol/L just before 18F-FDG intravenous injection, because a glucose level that is greater than 200 mg/dL might alter the FDG biodistribution. The dose of 18F-FDG ranges from 350–400 MBq (9.5–11 mCi) based on the body weight of patients ( $3.0\sim3.7$  MBq/kg) with an uptake period of 1 h, wherein the patient is informed to rest calmly.

## 3.4.2. PET/CT Scan Acquisition

During examinations, patients are comfortably placed on the PET/CT scanner for about 30 to 60 min (depending on the number of minutes per bed position and the field of view). After 60 min, the patient is typically imaged from the skull base to mid-thigh level. Noncontrast CT images are gained for attenuation correction and better anatomical localization.

After the imaging process, reconstruction of the CT images is carried out. The reconstructed CT images can then be used to generate an attenuation map to correct the attenuation of the PET images. Subsequently, a fusion between the corrected PET images and the CT images is performed to allow for the PET/CT images evaluation. All images are displayed in multiplanar reconstructions.

## 3.4.3. Qualitative Image Analysis

The CT component images allow qualitative interpretation by visual analysis of the lesion morphology. Nodules with irregular/spiculated margins, without typical benign calcification (popcorn-like, laminated, central nodular, and diffuse), and showing interval growth are considered malignant [3].

## 3.4.4. Quantitative Image Analysis

A standardized uptake value (SUV) is employed to semiquantitatively interpret the PET scans. SUV signifies the level of 18F-FDG uptake. If any lesion has an SUV higher than the background mediastinal blood pool uptake, then it is deemed hypermetabolic, and is expected to be malignant (Figure 2). Early studies of PET/CT in the literature approved 2.5 as a threshold SUVmax [49–51].



**Figure 2.** A 65-year-old male cigarette smoker with bronchogenic large-cell carcinoma. Axialfused FDG PET/CT image (**A**) reveals FDG-avid large left upper lung lobe mass ( $5.5 \times 6$  cm) with maximum standardized uptake value (SUVmax) of 11.8. Axial-fused FDG PET/CT image at distal level (**B**) reveals FDG-avid ipsilateral hilar LN with SUVmax 7. No FDG-avid distant metastatic deposits, imaging-based staging IIIA (T3 N1 M0).

#### 3.4.5. PET/CT Imaging Pitfalls

Increased uptake of 18F-FDG can be detected in certain benign lesions with an elevated intensity of inflammatory cells, such as granulomas and bacterial or fungal infections, that could likely result in finding false positives [52,53].

In addition, the use of single-time-point (STP) PET/CT is less specific in identifying malignancy in inhabitants with endemic infectious pulmonary disease in comparison with nonendemic regions. Thus, dual-time-point (DTP) is proposed, using retention index (RI) to enhance the diagnosis specificity of the PET/CT at granuloma-endemic regions with accurate distinction of benign and malignant SPNs [4].

## 3.4.6. Dual-Time-Point FDG PET/CT

The 18F-FDG uptake is routinely measured by semiquantitative SUVs, but some overlap in SUVs has been stated between begin and malignant nodules [54], so to overcome this diagnostic challenge, a DTP FDG-PET imaging technique can be performed to discriminate benign from malignant pulmonary lesions [55]. Malignant lesions with strong glycolysis and little or no glucose-6-phosphatase have decay-corrected SUVs on DTP imaging which progressively rise over time. This allows for improved contrast on delayed acquisition, compared to the nonincreasing SUVs of the neighboring normal tissue, due to certain glucose-6-phosphatase activity, for example, in the liver [56].

## 3.4.7. DTP FDG PET/CT Scan Acquisition

The process starts with a low-dosage CT scan with the following scanning parameters; a 3.75 mm slice thickness, a 1.75 mm pitch, 120 keV, and auto mA (35–100 mA, according to the body mass of the patient). Immediately after that, emission scanning is conducted without requiring the patient to change positions. Examinations of the whole body take approximately 1 hour (58 min  $\pm$  5) following the injection of 18F-FDG (194.5 MBq  $\pm$  27.3). In contrast, the delayed chest examinations are obtained in nearly 2 hours (120 min  $\pm$  4) following the injection of the 18F-FDG. The average time interval between early and delayed images is 60 min  $\pm$  6. The acquired attenuation-corrected CT data are then reconstructed utilizing a three-dimensional ordered-subset expectation maximization algorithm (image matrix size, 192 × 192; 16 subsets, 2 iterations: VUE Point Plus).

However, few of the DTP imaging studies found different sensitivity/specificity compared with STP imaging, with some positive and some negative findings [57,58]. Thus, sparse recent studies investigated the quantification of heterogeneity derived from various texture characteristics on DTP 18F-FDG PET/CT images and revealed that machine learning models trained by texture features accomplished substantial advances, compared to ordinary clinical metrics and visual assessment, in differentiating between benign and malignant SPNs, particularly by texture features on the delayed PET/CT images [4].

## 3.5. Magnetic Resonance Imaging (MRI)

MRI is of limited clinical use for lung disease patients, given that it compares poorly with other modalities such as CT, radiography, and PET/CT. Obtaining MRI of diagnostic quality of the lungs is very challenging due to many factors, including the low signal-to-noise ratio resulting from the lung's inherent low proton density, respiratory and cardiac motion artifacts, and susceptibility artifacts at the interface between tissue and air [59]. Lung imaging quality has improved by using a combination of free breathing, breath-hold, and gated protocols [60].

Pulmonary MRI is a perfect modality for patients demanding sequential follow-up, pediatric examinations, and pregnant women owing to its lack of ionizing radiation [61]. Recent MRI techniques, such as three-dimensional gradient sequences with ultrashort echo time and zero echo time using multicoil parallel acquisitions and acceleration techniques, have improved the small lung nodule detection in MRI [62]. Additionally, recent research has revealed the potential of using MRI for lung cancer screening, compared to low-dose CT [63].

The recently published Fleischner Society position paper considered pulmonary nodule characterization and lung cancer staging as an indication for current clinical use of pulmonary MRI [61].

## 3.6. Diffusion-Weighted Imaging (DWI)

Diffusion-weighted imaging (DWI) was found to be the most beneficial modality for pulmonary nodule characterization compared to other MRI assessed sequences. Metaanalysis reported a sensitivity of 83% and a specificity of 80% in distinguishing benign from malignant tumors [64]. DWI is a noncontrast functional MR sequence that uses the restriction of water molecule motion to assess tissue cellularity. Malignant lung tumors reveal restricted diffusion owing to their increased cellularity and reduced extracellular space [65] (Figure 3). The apparent diffusion coefficient (ADC) map can be used to perform qualitative and quantitative DWI analysis. Malignant lung tumors have a significantly low ADC compared to benign lesions (Figure 3). Thus, DWI is considered a radiation-free alternative for characterizing pulmonary nodules [66,67].

PET/CT is very expensive, with challenging assessment in some conditions because of false-negative results for well-differentiated pulmonary adenocarcinomas and small sizes of metabolically active tumors [68], and false-positive results for inflammatory nodules [52]. A few studies comparing DWI and PET/CT to distinguish among benign and malignant pulmonary nodules found that the two imaging modalities had similar accuracies [69–72]. A recent study discussed the value of combined assessment of DWI and T2-weighted imaging (T2WI) for discrimination of lung cancer from benign pulmonary nodules and masses and revealed that this combined assessment improves the diagnostic capability [2].

Another study revealed that diagnostic performance of DWI is comparable to, or even better than, that of PET/CT for classification of pulmonary lesions as malignant or benign [42]. Meta-analysis showed that DWI can discriminate between malignant and benign pulmonary lesions [73]. In addition, DWI is useful for assessing the N factor of lung cancer, with high diagnostic potential for nodal evaluation in NSCLC [74,75].



**Figure 3.** A 38-year-old male with bronchogenic adenocarcinoma. Axial T2 fast spin echo (FSE) with fat suppression (**A**) reveals spiculated left upper lung lobe mass of intermediate to high signal intensity. Diffusion-weighted image, high b-value =  $800 \text{ mm}^2/\text{s}$  (**B**) and corresponding ADC image (**C**) reveal a mass-constrained diffusion pattern with low ADC value of  $1.12 \times 10^{-3} \text{ mm}^2/\text{s}$ .

#### 3.6.1. DW Imaging Protocol

MR images can be acquired using a 1.5 T or 3 T magnetic machine with two front coils of six-channels body-phased array and two backward spinal clusters. Conventional MR images are performed before DWI and involve an axial T1-weighted spin-echo sequence and coronal and axial T2-weighted fast spin-echo.

DWIs are acquired using a single-shot echo-planar method with a 5 mm slice thickness under spectral attenuated inversion recovery (SPAIR) with a respiratory triggered scan using diffusion gradient encoding in three orthogonal directions with variable b-values, including b-values = 0 and 800 s/mm<sup>2</sup> [2,67], b-values = 0 and 500 s/mm<sup>2</sup> [76], or b-values = 20, 500, and 800 s/mm<sup>2</sup> [77].

Matched ADC maps are generated using at least two images of different b-values using a workstation with functional tool software.

## 3.6.2. Qualitative Image Analysis

Restricted diffusion is considered as visually detected hyperintense signal on DWI at high b-values of 500 and 800 s/mm<sup>2</sup> and isointense or hypointense signal on the ADC map compared to adjacent, normal lung (Figure 3).

#### 3.6.3. Quantitative Image Analysis

ADC values are measured by manually applying a 2D circular or oval region of interest (ROI) centrally to the lesion on ADC map, referring to T2-weighted, occupying as much of the lesion as possible (at least 50%) without including the adjacent normal lung. The ADC values are measured three times, then the measurements are averaged.

In addition, other DWI parameters, including signal intensity (SI) score and lesion to spinal cord signal intensity ratio (LSR), are feasible in distinction between benign and malignant SPNs [76].

#### 3.6.4. Intravoxel Incoherent Motion (IVIM)

An intravoxel incoherent motion (IVIM) study is a modification of DWI protocol where images are acquired at lower than conventional b-values, which are more sensitive to blood microcirculation. The IVIM model can differentiate between lung cancer and consolidation as compared to PET–CT [78]. Three parameters can be gained from IVIM, including the true diffusion coefficient (D value), perfusion-related pseudodiffusion coefficient (D\* value), and perfusion fraction (f value) [79]. Ki-67 is a nuclear antigen commonly applied as a tumor cell proliferation marker in standard pathologic analyses for lung cancer [80]. A meta-analysis showed that high Ki-67 expression is correlated with disease progression and poor outcome in lung cancer patients [81], although the assessment of Ki-67 expression using immunohistochemistry (IHC) is invasive and is often affected by tissue sample quality [82]. A recent study compared the quantitative MRI parameters from conventional DWI, IVIM, and diffusion kurtosis imaging (DKI) in correlations with tumor Ki-67 and revealed that the (D value) from IVIM displayed the best performance among the three techniques for differentiating the high-Ki-67 group from the low-Ki-67 group, as well as distinguishing SCLC from NSCLC [83].

#### 3.7. Dynamic Contrast-Enhanced (DCE) MRI

The dynamic contrast-enhanced (DCE) MRI with 2D or 3D SE, turbo SE, and/or GRE sequences has been utilized for differentiating malignant from benign pulmonary nodules with a sensitivity, specificity, and accuracy ranging from 55% to 100%, 54% to 100%, and 75% to 96%, respectively [84–86]. Though enhancement levels may differ owing to underlying pathologic conditions such as tumor angiogenesis, the presence or absence of fibrosis, and necrosis inside the tumor, malignant pulmonary nodules display homogenous enhancement, but at variable levels on T1-weighted images following contrast media administration [84,87]. DCE MRI also provides signal intensity/time curves which represent first-transit and washout of contrast media under breath-hold or repeated breathhold conditions for 5 min. The enhancement patterns within nodules on DCE MRI and/or parameters determined using signal intensity/time curves can be assessed to differentiate between benign and malignant nodules. Many studies indicated that enhancement patterns assessed with DCE MRI may be beneficial for the diagnosis and management of pulmonary nodules [84,85]. A meta-analysis reported no significant differences in the diagnostic performance of CECT, DCE MRI, and FDG PET in the evaluation of pulmonary nodules [88]. Consequently, DCE MRI has a corresponding role in CT and FDG PET/CT in establishing the necessity for additional intervention and treatment in routine clinical practice [87].

## 3.8. Hyperpolarized gas MRI

Lung diseases impact the airways, the gas exchange parenchyma, or both, with significant influence on patients' morbidity [89]. Hyperpolarized (HP) gas MR imaging using helium (3He) and xenon (129Xe) offers a noninvasive, ionizing radiation–free quantitative procedure to assess pulmonary structure, gas exchange, and function [90]. Owing to the paucity in the supply of helium and its high price, the less expensive and organically available xenon has been adopted in HP gas MRI. Previous study revealed a substantial quantitative agreement for ventilation maps on hyperpolarized 3He MRI and 4DCT imaging in patients with lung cancer [91]. Additionally, HP gas MRI is helpful in radiotherapy planning for lung cancer, utilizing functional information provided by HP gas MRI that has been registered for treatment planning.CT revealed statistically significant improvements to radiotherapy planning, as well as the mean lung dose to the functional lung [92].

## 3.9. Whole-Body Magnetic Resonance Imaging (WB-MRI) and WB-DWI

Whole-body magnetic resonance imaging (WB-MRI) is a non-invasive modality for cancer staging with better diagnostic accuracy after inclusion of DWI [93]. Recent metaanalysis revealed that WB-MRI and WB-DWI are radiation-free alternatives that have diagnostic performance similar to that of 18F-FDG PET/CT in M-staging of NSCLC with reduced cost and time compared to PET/CT [94].

#### PET/MRI

Lately, hybrid PET/MRI has been involved in abdominal or pelvic cancer staging due to the opportunity to combine multiparametric metabolic, morphological, and functional data provided by radioactive tracers and various MRI sequences [95].

PET/MRI has also become popular for evaluating NSCLC and pulmonary lesions, but owing to the diagnostic challenges of the MRI lung, pulmonary nodule size is an important factor to assess PET/MRI diagnostic performance [96]. Adjusting large FOV DWI is critical for PET/MRI, especially with regard to the characterization of pulmonary nodules, and recent advances involving usage of reverse-phase encoding with DWI could assist in providing a quantitative advantage to PET/MRI [77].

Recent work analyzed the available studies on PET/MRI for staging in NSCLC or lung nodule and revealed that PET/MRI had similar sensitivity for detecting FDG-avid lung nodules and nodules greater than 10 mm, but PET/CT attained a greater detection rate in non-FDG-avid lung nodules below 5 mm. In addition, when compared to PET/CT, PET/MRI had no advantages in T- and N-staging of NSCLC [96].

## 4. Imaging and Lung Cancer TNM Staging

The latest (eighth) edition of TNM Staging System, released in January 2017, is currently followed in clinical practice for lung cancer staging [97].

#### 4.1. T (Tumor) Descriptor

The scope of spread of the primary tumor is shown by the T descriptors, consisting of tumor size, invasion, and site of separate tumor in relation to the primary tumor [98]. Tumor size must be quantified with CT utilizing the lung window, and the largest dimension in any plane has to be stated [99].

The cutoff points of 3 cm, 5 cm, and 7 cm separate T1 from T2 tumors, T3 from T2, and T3 from T4, respectively. The eighth edition of lung cancer staging announced a new T category of Tis and T1mi. Tis now is allocated for carcinoma in situ of squamous cell carcinoma and adenocarcinoma, unlike the seventh edition, where Tis was only employed for squamous cell carcinoma in situ. The T descriptors are now sorted into five main categories—see Table 1.

The T classification is modified in the eighth edition of TNM lung cancer staging built on 1 cm increments in tumor size, downstage of endobronchial tumor carina to T2, integrating atelectasis or pneumonitis into T2, and considering the diaphragmatic invasion as T4 [98].

Commonly, CT well illustrates the invasion of the mediastinal structures and great vessels, and transthoracic ultrasound and MRI assess chest wall and pleural invasion better than CT [100]. For Pancoast tumors, MRI demonstrated better results than CT in preoperative staging [101].

T Descriptor			
Tis (AIS)	Pure GGN is $\leq$ 3.0 cm.		
T1mi	Less than or equal to 0.5 cm solid part in tumor total size $\leq$ 3.0 cm.		
T1a	0.6–1.0 cm solid part in tumor size $\leq$ 3.0 cm. Pure GGN is greater than 3 cm. $\leq$ 1 cm solid tumor.		
T1b	1.1–2.0 cm solid part in tumor size less than or equal to 3.0 cm. Greater than 1.0–2.0 cm solid tumor.		
T1c	2.1–3.0 cm solid part in tumor size less than or equal to 3.0 cm. Greater than 2.0–3.0 cm solid tumor.		
T2a	3.1–4.0 cm. Invades main bronchus (no carinal involvement).		
T2b	4.1–5.0 cm. Total/partial atelectasis, pneumonitis. Involves visceral pleura (PL1 or PL2).		
Τ3	5.1–7.0 cm. Tumor nodule in the same lobe as the primary tumor. Directly invades any of the following: chest wall, parietal pleura (PL3), parietal pericardium, or phrenic nerve.		
T4	Greater than 7.0 cm. Tumor nodule in different ipsilateral lobe than that of primary tumor. Directly invades any of the following: diaphragm, mediastinum, tra- chea, carina, great vessels, heart, recurrent laryngeal nerve, esophagus, or vertebral body.		
N Descriptor			
N0	No LN metastasis.		
N1	Metastasis to ipsilateral peribronchial, intrapulmonary, or hilar LNs.		
N2	Metastasis to ipsilateral mediastinal or subcarinal LNs.		
N3	Metastasis to ipsilateral or contralateral supraclavicular/scalene LNs. Metastasis to contralateral mediastinal, hilar LNs.		
M Descriptor			
M0	No distant metastasis.		
M1a	Malignant pleural effusion or pericardial effusion. Contralateral lung nodules/pleural nodules.		
M1b	Single extrathoracic metastasis.		
M1c	Multiple extrathoracic metastasis.		

Table 1. Imaging appearance for TNM lung cancer staging in the eighth edition [98,101].

GGN, AIS, mi, PL1, PL2, PL3, and LN refer to ground-glass nodules, adenocarcinoma in situ, minimally invasive, tumor that reaches the elastic layer of visceral pleura without extending to its surface, tumor that reaches the surface of the visceral pleural, tumor that reaches the parietal pleura or chest wall, and lymph node, respectively [98,102,103].

## 4.2. N (Nodal) Descriptor

The classification of nodal metastasis is established on structural position of affected lymph nodes (LNs), unlike other cancer staging which often depend on the number of metastatic LNs. The N descriptor categorization of the seventh edition remained the same for the eighth edition, classified into four main categories (Table 1) [104].

The classic criteria for diagnosis of LN involvement is LN greater than 10 mm in short axis [102]. In the diagnosis of LN involvement, PET–CT revealed better outcomes than CT [105]. DWI can discriminate between benign and malignant LNs, as a meta-analysis revealed that diffusion MRI has the same sensitivity as PET–CT but has higher specificity than PET–CT [106], and other trials showed comparable diagnostic performance of PET–CT and DWI [107].

## 4.3. M (Metastasis) Descriptor

The M descriptors in the eighth edition for M1a intrathoracic metastasis had no significant change from the seventh edition [98]. Extrathoracic metastasis in the eighth edition was subdivided into M1b and M1c, depending on the number of metastatic lesions, due to significantly poor prognosis for those with multiple extrathoracic metastasis (M1c) (Table 1) [102].

Lung cancer frequently spreads to the adrenal glands, brain, and bone marrow [108], so an MRI provides more image contrast flexibility than CT. Cerebral staging is advised in patients with promising curative opportunities, as MRI outperforms CT in the detection of brain metastases [109]. Adrenal metastases are accurately detected by PET–CT [110]. Generally, PET–CT is the best for initial lung cancer staging compared to bone and CT scans, except for brain metastases [101].

## 4.4. Overall Stage Grouping

Due to updates in the T and M descriptors, adjustments of current staging categories and creation of new staging groups were designed for the eighth edition (Table 2) [98].

Stage	Μ	Ν	Т
0	M0	N0	Tis
IA1	M0	N0	T1mi
	M0	N0	T1a
IA2	M0	N0	T1b
IA3	M0	N0	T1c
IB	M0	N0	T2a
IIA	M0	N0	T2b
IIB	M0	N1	T1a, b, c
	M0	N1	T2a, b
	M0	N0	T3
IIIA	M0	N2	T1a, b, c
	M0	N2	T2a, b
	M0	N1	Т3
	M0	N0	T4
	M0	NI	T4
IIIB	M0	N3	T1a, b, c
	M0	N3	T2a, b
	M0	N2	Т3
	M0	N2	Τ4
IIIC	M0	N3	T3
	M0	N3	T4
IVA	M1a	Any N	Any T
	M1b	Any N	Any T
IVB	M1c	Any N	Any T

Table 2. Overall stage grouping of TNM lung cancer in the eighth edition [98,101].

## 5. Lung Cancer Follow-Up and Response Evaluation

Surgery is typically the primary treatment modality for patients with localized disease. Post-treatment follow-up comprises imaging, physical examination, and laboratory tests combinations. There is no individual modality that can be sensitive, specific, and cost-effective at the same time; therefore, a mixed approach is required to detect tumor recurrence [46]. Proper imaging is a key model for response evaluation of lung cancer and is essential to identify progression of the disease during chemotherapy with cytotoxic agents, molecular targeting therapy, and blend therapy of both types [111].

## 6. World Health Organization (WHO) Criteria and Response Evaluation Criteria in Solid Tumors (RECIST)

Standardized tumor response criteria, such as WHO criteria and RECIST, are simple reasonable measures for response assessment through institutes [111]. These are mainly based on size, involving bidimensional tumor measurements. In the WHO criteria, these are the largest perpendicular dimensions in the axial direction, while RECIST uses the unidimensional diameter, which is the largest tumor dimension in the axial plane [111,112].

Though both WHO and RECIST criteria are broadly used in routine practice and clinical trials for assessing treatment response, they have a few limitations. Only maximum tumor diameters are employed for analysis without functional information [113]. Interreader variability has been reported for tumor size measurements, especially in irregular lesions, and thus may tend to misclassify tumor progression. In addition, the size measurements of lung cancer do not precisely discriminate a residual tumor from post-radiotherapy changes [46]. Additionally, the CT cannot reliably distinguish a histopathologic response (described as less than or equal to 10% viable tumor). After neoadjuvant chemotherapy, a discordance rate of 41% between histopathologic response and CT-based RECIST response classification was reported in patients with NSCLC [114].

FDG PET/CT is a potent tool in post-treatment response evaluation, and it predicts survival better than CT, as FDG PET/CT increases accuracy and prediction of disease progression since cellular metabolism changes arise more quickly than tumor size changes [115]. Criteria for assessing PET response in tumors have been presented to be a stronger predictor of pathologic response than anatomic response indicators, as WHO criteria, RECIST, and PET/CT detect the response by measuring the maximum SUV in the tumor [112].

The NCCN advised FDG PET/CT for proper staging and accurate radiation therapy (RT) planning for both NSCLC and SCLC, but the NCCN does not recommend the routine use of FDG PET/CT for assessment and follow-up treatment response [46].

Various studies demonstrated the benefit of post-treatment FDG PET/CT SUVmax to predicttreatment response and prognosis, usually achieved within 6 months after treatment completion [46,116]. Additionally, meta-analysis stated that the measured SUVmax in the tumor before and after RT was competent to predictpatient prognosis [117].

DWI functional assessment was of more value than CT in evaluating response after chemotherapy and/or radiotherapy in lung cancer [113]. DWI is comparable and could have a greater chance of predicting tumor response to therapy in NSCLC patients before chemoradiotherapy than FDG PET/CT. Both ADC and SUVmax values are accepted as radiologic examination-based biomarkers [118]. ADC measurements are a promising way to monitor early response or predict prognosis after NSCLC chemotherapy [119,120].

#### 7. Advances in Lung Cancer Tumor Genomics and Precision Therapy

Recent discoveries in the characterization of tumor genomic abnormalities and effective precision-targeted therapy for certain genomic aberrations in lung cancer achieved innovative progress in lung cancer treatment and altered the treatment methodologies for lung cancer in the last decade. Precision therapy has been brought to the focus of lung cancer therapy [111,121]. Radiologists must keep up with the revolutionary world of genomic alterations and the rapidly evolving therapy with anticancer agents that target these genomic alterations in order to maintain effective communication with oncologists in multidisciplinary teams for the care of patients [111,121].

Somatic activating mutations of the tyrosine kinase domain of the epidermal growth factor receptor (EGFR) gene are a representative example of genomic alterations in NSCLC. They respondwell to EGFR tyrosine kinase inhibitors (EGFR-TKI), gefitinib, and erlotinib target therapy [122,123]. EGFR mutations are the most common in women with ade-nocarcinoma who have no history of cigarette smoking and are less common in elderly heavy-smoker patients [124].

The rearranged anaplastic lymphoma kinase (*ALK*) oncogene is another example of genomic alterations in NSCLC and adenocarcinoma. It is more common in women, young patients, non-smokers, or light smokers [125].

EGFR and *ALK* mutation testing is suggested for patients with NSCLC, adenocarcinoma, and large-cell carcinoma by the National Comprehensive Cancer Network Clinical Practice Guidelines in Oncology. If both are negative, chemotherapy is started. If positive for EGFR mutation, erlotinib target therapy is started; if positive for *ALK* mutation, crizotinib target therapy is started. Patients who suffer from squamous cell carcinoma are not recommended to undergo EGFR and *ALK* mutation testing, and should instead proceed to chemotherapy [126].

The precision therapy strategies for lung cancer are rapidly advancing, enabling the introduction of novel methods and genomic maps analyses into the clinical practice. The radiologist's knowledge of these recent therapeutic plans and their relation to the pre- and post-treatment imaging become essential, especially among members of multidisciplinary lung cancer teams [121].

## 8. Conclusions

Updated knowledge of lung cancer genomic abnormalities, advanced imaging techniques, treatment regimens, and tumor response assessment helps the radiologists to continue as effective contributors to the customized care of lung cancer patients. Details of lung cancer classifications and different state-of-the-art screening techniques are presented in this review to serve as a source of learning and research.

Author Contributions: Conceptualization, N.M.B., G.A.S., A.S. (Ahmed Sharafeldeen), H.K., A.M., A.S. (Ahmed Shalaby), M.Y., A.K., M.G. and A.E.-B.; project administration, A.E.-B.; writing—original draft, N.M.B., G.A.S., A.S. (Ahmed Sharafeldeen), H.K., A.M., A.S. (Ahmed Shalaby), M.Y., A.K., M.G. and A.E.-B.; writing—review and editing, N.M.B., G.A.S., A.S. (Ahmed Sharafeldeen), H.K., A.M., A.S. (Ahmed Sharafeldeen), H.K., A.M., A.S. (Ahmed Sharafeldeen), H.K., A.M., M.G. and A.E.-B.; writing—review and editing, N.M.B., G.A.S., A.S. (Ahmed Sharafeldeen), H.K., A.M., A.S. (Ahmed Sharafeldeen), H.K., A.M., A.S. (Ahmed Sharafeldeen), H.K., P. (Ahmed Sharafeldeen), H.K., A.M., A.S. (Ahmed Sharafeldeen), H.K., P. (Almed Sharafeldeen), H.K., A.M., A.S. (Ahmed Sharafeldeen), H.K., A.K., M.G. a

**Funding:** This research is supported by Abu Dhabi's Advanced Technology Research Council via the ASPIRE Award for Research Excellence program.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare that they have no competing interests.

## References

- 1. Siegel, R.L.; Miller, K.D.; Jemal, A. Cancer statistics, 2015. CA Cancer J. Clin. 2015, 65, 5–29. [CrossRef]
- Usuda, K.; Ishikawa, M.; Iwai, S.; Iijima, Y.; Motono, N.; Matoba, M.; Doai, M.; Hirata, K.; Uramoto, H. Combination Assessment of Diffusion-Weighted Imaging and T2-Weighted Imaging Is Acceptable for the Differential Diagnosis of Lung Cancer from Benign Pulmonary Nodules and Masses. *Cancers* 2021, 13, 1551. [CrossRef] [PubMed]
- Sim, Y.T.; Goh, Y.G.; Dempsey, M.F.; Han, S.; Poon, F.W. PET–CT Evaluation of Solitary Pulmonary Nodules: Correlation with Maximum Standardized Uptake Value and Pathology. *Lung* 2013, 191, 625–632. [CrossRef] [PubMed]

- 4. Chen, S.; Harmon, S.; Perk, T.; Li, X.; Chen, M.; Li, Y.; Jeraj, R. Diagnostic classification of solitary pulmonary nodules using dual time 18F-FDG PET/CT image texture features in granuloma-endemic regions. *Sci. Rep.* **2017**, *7*, 9370. [CrossRef] [PubMed]
- 5. American College of Radiology. Lung-RADS Version 1.0 Assessment Categories. 2014. Available online: https://www.uoflhealthnetwork.org/documents/lung-rads-categories.pdf (accessed on 4 April 2022).
- Martin, M.D.; Kanne, J.P.; Broderick, L.S.; Kazerooni, E.A.; Meyer, C.A. Lung-RADS: Pushing the Limits. *RadioGraphics* 2017, 37, 1975–1993. [CrossRef]
- American College of Radiology. Lung-RADS Version 1.1. 2019. Available online: https://www.acr.org/-/media/ACR/Files/ RADS/Lung-RADS/LungRADSAssessmentCategoriesv1-1.pdf (accessed on 4 April 2022).
- Kastner, J.; Hossain, R.; Jeudy, J.; Dako, F.; Mehta, V.; Dalal, S.; Dharaiya, E.; White, C. Lung-RADS Version 1.0 versus Lung-RADS Version 1.1: Comparison of Categories Using Nodules from the National Lung Screening Trial. *Radiology* 2021, 300, 199–206. [CrossRef]
- MacMahon, H.; Austin, J.H.M.; Gamsu, G.; Herold, C.J.; Jett, J.R.; Naidich, D.P.; Patz, E.F.; Swensen, S.J. Guidelines for Management of Small Pulmonary Nodules Detected on CT Scans: A Statement from the Fleischner Society. *Radiology* 2005, 237, 395–400. [CrossRef]
- Naidich, D.P.; Bankier, A.A.; MacMahon, H.; Schaefer-Prokop, C.M.; Pistolesi, M.; Goo, J.M.; Macchiarini, P.; Crapo, J.D.; Herold, C.J.; Austin, J.H.; et al. Recommendations for the Management of Subsolid Pulmonary Nodules Detected at CT: A Statement from the Fleischner Society. *Radiology* 2013, 266, 304–317. [CrossRef]
- MacMahon, H.; Naidich, D.P.; Goo, J.M.; Lee, K.S.; Leung, A.N.C.; Mayo, J.R.; Mehta, A.C.; Ohno, Y.; Powell, C.A.; Prokop, M.; et al. Guidelines for Management of Incidental Pulmonary Nodules Detected on CT Images: From the Fleischner Society 2017. *Radiology* 2017, 284, 228–243. [CrossRef]
- 12. Panunzio, A.; Sartori, P. Lung Cancer and Radiological Imaging. Curr. Radiopharm. 2020, 13, 238–242. [CrossRef]
- Tsao, M.S.; Nicholson, A.G.; Maleszewski, J.J.; Marx, A.; Travis, W.D. Introduction to 2021 WHO Classification of Thoracic Tumors. J. Thorac. Oncol. 2022, 17, e1–e4. [CrossRef]
- Chen, Z.; Fillmore, C.M.; Hammerman, P.S.; Kim, C.F.; Wong, K.K. Non-small-cell lung cancers: A heterogeneous set of diseases. *Nat. Rev. Cancer* 2014, 14, 535–546. [CrossRef]
- 15. Davidson, M.R.; Gazdar, A.F.; Clarke, B.E. The pivotal role of pathology in the management of lung cancer. *J. Thorac. Dis.* **2013**, *5*, S463.
- Han, Y.; Ma, Y.; Wu, Z.; Zhang, F.; Zheng, D.; Liu, X.; Tao, L.; Liang, Z.; Yang, Z.; Li, X.; et al. Histologic subtype classification of non-small cell lung cancer using PET/CT images. *Eur. J. Nucl. Med. Mol. Imaging* 2020, 48, 350–360. [CrossRef]
- 17. Purandare, N.C.; Rangarajan, V. Imaging of lung cancer: Implications on staging and management. *Indian J. Radiol. Imaging* 2015, 25, 109–120. [CrossRef]
- Forster, B.B.; Müller, N.L.; Miller, R.R.; Nelems, B.; Evans, K.G. Neuroendocrine carcinomas of the lung: Clinical, radiologic, and pathologic correlation. *Radiology* 1989, 170, 441–445. [CrossRef]
- 19. Grant, L.A.; Griffin, N. Grainger & Allison's Diagnostic Radiology Essentials E-Book; Elsevier Health Sciences: London, UK, 2018.
- Owonikoko, T.K.; Ragin, C.C.; Belani, C.P.; Oton, A.B.; Gooding, W.E.; Taioli, E.; Ramalingam, S.S. Lung Cancer in Elderly Patients: An Analysis of the Surveillance, Epidemiology, and End Results Database. J. Clin. Oncol. 2007, 25, 5570–5577. [CrossRef]
- Okubo, K.; Mark, E.J.; Flieder, D.; Wain, J.C.; Wright, C.D.; Moncure, A.C.; Grillo, H.C.; Mathisen, D.J. Bronchoalveolar carcinoma: Clinical, radiologic, and pathologic factors and survival. *J. Thorac. Cardiovasc. Surg.* 1999, 118, 702–709. [CrossRef]
- de Hoop, B.; Schaefer-Prokop, C.; Gietema, H.A.; de Jong, P.A.; van Ginneken, B.; van Klaveren, R.J.; Prokop, M. Screening for Lung Cancer with Digital Chest Radiography: Sensitivity and Number of Secondary Work-up CT Examinations. *Radiology* 2010, 255, 629–637. [CrossRef]
- Sone, S.; Li, F.; Yang, Z.G.; Takashima, S.; Maruyama, Y.; Hasegawa, M.; Wang, J.C.; Kawakami, S.; Honda, T. Characteristics of small lung cancers invisible on conventional chest radiography and detected by population based screening using spiral CT. *Br. J. Radiol.* 2000, 73, 137–145. [CrossRef]
- Thakur, S.K.; Singh, D.P.; Choudhary, J. Lung cancer identification: A review on detection and classification. *Cancer Metastasis Rev.* 2020, 39, 989–998. [CrossRef] [PubMed]
- Meisinger, Q.C.; Klein, J.S.; Butnor, K.J.; Gentchos, G.; Leavitt, B.J. CT Features of Peripheral Pulmonary Carcinoid Tumors. Am. J. Roentgenol. 2011, 197, 1073–1080. [CrossRef] [PubMed]
- Saito, H.; Yamada, K.; Hamanaka, N.; Oshita, F.; Ito, H.; Nakayama, H.; Yokose, T.; Kameda, Y.; Noda, K. Initial Findings and Progression of Lung Adenocarcinoma on Serial Computed Tomography Scans. J. Comput. Assist. Tomogr. 2009, 33, 42–48. [CrossRef] [PubMed]
- Swensen, S.J.; Viggiano, R.W.; Midthun, D.E.; Müller, N.L.; Sherrick, A.; Yamashita, K.; Naidich, D.P.; Patz, E.F.; Hartman, T.E.; Muhm, J.R.; et al. Lung Nodule Enhancement at CT: Multicenter Study. *Radiology* 2000, 214, 73–80. [CrossRef] [PubMed]
- 28. Huo, J.W.; Huang, X.T.; Li, X.; Gong, J.W.; Luo, T.Y.; Li, Q. Pneumonic-type lung adenocarcinoma with different ranges exhibiting different clinical, imaging, and pathological characteristics. *Insights Imaging* **2021**, *12*, 169. [CrossRef] [PubMed]
- McCollough, C.H.; Leng, S.; Yu, L.; Fletcher, J.G. Dual- and Multi-Energy CT: Principles, Technical Approaches, and Clinical Applications. *Radiology* 2015, 276, 637–653. [CrossRef]
- Sellerer, T.; Noël, P.B.; Patino, M.; Parakh, A.; Ehn, S.; Zeiter, S.; Holz, J.A.; Hammel, J.; Fingerle, A.A.; Pfeiffer, F.; et al. Dual-energy CT: A phantom comparison of different platforms for abdominal imaging. *Eur. Radiol.* 2018, 28, 2745–2755. [CrossRef]

- Kim, H.; Goo, J.M.; Kang, C.K.; Chae, K.J.; Park, C.M. Comparison of Iodine Density Measurement Among Dual-Energy Computed Tomography Scanners From 3 Vendors. *Investig. Radiol.* 2018, 53, 321–327. [CrossRef]
- Deniffel, D.; Sauter, A.; Dangelmaier, J.; Fingerle, A.; Rummeny, E.J.; Pfeiffer, D. Differentiating intrapulmonary metastases from different primary tumors via quantitative dual-energy CT based iodine concentration and conventional CT attenuation. *Eur. J. Radiol.* 2019, 111, 6–13. [CrossRef]
- Chen, X.; Xu, Y.; Duan, J.; Li, C.; Sun, H.; Wang, W. Correlation of iodine uptake and perfusion parameters between dual-energy CT imaging and first-pass dual-input perfusion CT in lung cancer. *Medicine* 2017, 96, e7479. [CrossRef]
- Son, J.Y.; Lee, H.Y.; Kim, J.H.; Han, J.; Jeong, J.Y.; Lee, K.S.; Kwon, O.J.; Shim, Y.M. Quantitative CT analysis of pulmonary groundglass opacity nodules for distinguishing invasive adenocarcinoma from non-invasive or minimally invasive adenocarcinoma: The added value of using iodine mapping. *Eur. Radiol.* 2015, 26, 43–54. [CrossRef]
- Chae, E.J.; Song, J.W.; Krauss, B.; Song, K.S.; Lee, C.W.; Lee, H.J.; Seo, J.B. Dual-energy Computed Tomography Characterization of Solitary Pulmonary Nodules. *J. Thorac. Imaging* 2010, 25, 301–310. [CrossRef] [PubMed]
- 36. Altenbernd, J.; Wetter, A.; Umutlu, L.; Hahn, S.; Ringelstein, A.; Forsting, M.; Lauenstein, T. Dual-energy computed tomography for evaluation of pulmonary nodules with emphasis on metastatic lesions. *Acta Radiol.* **2015**, *57*, 437–443. [CrossRef] [PubMed]
- Baxa, J.; Vondráková, A.; Matoušková, T.; Růžičková, O.; Schmidt, B.; Flohr, T.; Sedlmair, M.; Ferda, J. Dual-phase dual-energy CT in patients with lung cancer: Assessment of the additional value of iodine quantification in lymph node therapy response. *Eur. Radiol.* 2014, 24, 1981–1988. [CrossRef]
- den Harder, A.M.; Bangert, F.; van Hamersvelt, R.W.; Leiner, T.; Milles, J.; Schilham, A.M.R.; Willemink, M.J.; de Jong, P.A. The Effects of Iodine Attenuation on Pulmonary Nodule Volumetry using Novel Dual-Layer Computed Tomography Reconstructions. *Eur. Radiol.* 2017, 27, 5244–5251. [CrossRef] [PubMed]
- Monnin, P.; Sfameni, N.; Gianoli, A.; Ding, S. Optimal slice thickness for object detection with longitudinal partial volume effects in computed tomography. J. Appl. Clin. Med Phys. 2017, 18, 251–259.
- Concatto, N.H.; Watte, G.; Marchiori, E.; Irion, K.; Felicetti, J.C.; Camargo, J.J.; Hochhegger, B. Magnetic resonance imaging of pulmonary nodules: Accuracy in a granulomatous disease–endemic region. *Eur. Radiol.* 2015, 26, 2915–2920. [CrossRef]
- 41. Wang, Y.X.J.; Lo, G.G.; Yuan, J.; Larson, P.E.Z.; Zhang, X. Magnetic resonance imaging for lung cancer screen. *J. Thorac. Dis.* **2014**, *6*, 1340.
- Dias, A.B.; Zanon, M.; Altmayer, S.; Pacini, G.S.; Concatto, N.H.; Watte, G.; Garcez, A.; Mohammed, T.L.; Verma, N.; Medeiros, T.; et al. Fluorine 18–FDG PET/CT and Diffusion-weighted MRI for Malignant versus Benign Pulmonary Lesions: A Meta-Analysis. *Radiology* 2019, 290, 525–534. [CrossRef]
- Nakajo, M.; Jinguji, M.; Aoki, M.; Tani, A.; Sato, M.; Yoshiura, T. The clinical value of texture analysis of dual-time-point 18F-FDG-PET/CT imaging to differentiate between 18F-FDG-avid benign and malignant pulmonary lesions. *Eur. Radiol.* 2019, 30, 1759–1769. [CrossRef]
- Gould, M.K.; Maclean, C.C.; Kuschner, W.G.; Rydzak, C.E.; Owens, D.K. Accuracy of Positron Emission Tomography for Diagnosis of Pulmonary Nodules and Mass Lesions. *JAMA* 2001, 285, 914. [CrossRef]
- Jeong, S.Y.; Lee, K.S.; Shin, K.M.; Bae, Y.A.; Kim, B.T.; Choe, B.K.; Kim, T.S.; Chung, M.J. Efficacy of PET/CT in the characterization of solid or partly solid solitary pulmonary nodules. *Lung Cancer* 2008, *61*, 186–194. [CrossRef] [PubMed]
- Sheikhbahaei, S.; Mena, E.; Yanamadala, A.; Reddy, S.; Solnes, L.B.; Wachsmann, J.; Subramaniam, R.M. The Value of FDG PET/CT in Treatment Response Assessment, Follow-Up, and Surveillance of Lung Cancer. *Am. J. Roentgenol.* 2017, 208, 420–433. [CrossRef]
- 47. van Velden, F.H.P.; Cheebsumon, P.; Yaqub, M.; Smit, E.F.; Hoekstra, O.S.; Lammertsma, A.A.; Boellaard, R. Evaluation of a cumulative SUV-volume histogram method for parameterizing heterogeneous intratumoural FDG uptake in non-small cell lung cancer PET studies. *Eur. J. Nucl. Med. Mol. Imaging* **2011**, *38*, 1636–1647. [CrossRef]
- Naqa, I.E.; Grigsby, P.; Apte, A.; Kidd, E.; Donnelly, E.; Khullar, D.; Chaudhari, S.; Yang, D.; Schmitt, M.; Laforest, R.; et al. Exploring feature-based approaches in PET images for predicting cancer treatment outcomes. *Pattern Recognit.* 2009, 42, 1162–1171. [CrossRef]
- 49. Lowe, V.J.; Fletcher, J.W.; Gobar, L.; Lawson, M.; Kirchner, P.; Valk, P.; Karis, J.; Hubner, K.; Delbeke, D.; Heiberg, E.V.; et al. Prospective investigation of positron emission tomography in lung nodules. *J. Clin. Oncol.* **1998**, *16*, 1075–1084. [CrossRef]
- Christensen, J.A.; Nathan, M.A.; Mullan, B.P.; Hartman, T.E.; Swensen, S.J.; Lowe, V.J. Characterization of the Solitary Pulmonary Nodule:18F-FDG PET Versus Nodule-Enhancement CT. Am. J. Roentgenol. 2006, 187, 1361–1367. [CrossRef] [PubMed]
- Hashimoto, Y.; Tsujikawa, T.; Kondo, C.; Maki, M.; Momose, M.; Nagai, A.; Ohnuki, T.; Nishikawa, T.; Kusakabe, K. Accuracy of PET for diagnosis of solid pulmonary lesions with 18F-FDG uptake below the standardized uptake value of 2.5. *J. Nucl. Med.* 2006, 47, 426–431.
- Goo, J.M.; Im, J.G.; Do, K.H.; Yeo, J.S.; Seo, J.B.; Kim, H.Y.; Chung, J.K. Pulmonary Tuberculoma Evaluated by Means of FDG PET: Findings in 10 Cases. *Radiology* 2000, 216, 117–121. [CrossRef]
- Igai, H.; Gotoh, M.; Yokomise, H. Computed tomography (CT) and positron emission tomography with [18F]fluoro-2-deoxy-dglucose (FDG-PET) images of pulmonary cryptococcosis mimicking lung cancer. *Eur. J. Cardio-Thorac. Surg.* 2006, 30, 837–839. [CrossRef]
- Nguyen, N.C.; Kaushik, A.; Wolverson, M.K.; Osman, M.M. Is there a common SUV threshold in oncological FDG PET/CT, at least for some common indications? A retrospective study. *Acta Oncol.* 2011, 50, 670–677. [CrossRef]

- 55. Matthies, A.; Hickeson, M.; Cuchiara, A.; Alavi, A. Dual time point 18F-FDG PET for the evaluation of pulmonary nodules. *J. Nucl. Med.* **2002**, *43*, 871–875.
- Parghane, R.V.; Basu, S. Dual-time point 18 F-FDG-PET and PET/CT for Differentiating Benign From Malignant Musculoskeletal Lesions: Opportunities and Limitations. *Semin. Nucl. Med.* 2017, 47, 373–391. [CrossRef]
- Suga, K.; Kawakami, Y.; Hiyama, A.; Sugi, K.; Okabe, K.; Matsumoto, T.; Ueda, K.; Tanaka, N.; Matsunaga, N. Dual-time point 18F-FDG PET/CT scan for differentiation between 18F-FDG-avid non-small cell lung cancer and benign lesions. *Ann. Nucl. Med.* 2009, 23, 427–435. [CrossRef]
- 58. Cloran, F.J.; Banks, K.P.; Song, W.S.; Kim, Y.; Bradley, Y.C. Limitations of dual time point PET in the assessment of lung nodules with low FDG avidity. *Lung Cancer* 2010, *68*, 66–71. [CrossRef]
- 59. Wild, J.M.; Marshall, H.; Bock, M.; Schad, L.R.; Jakob, P.M.; Puderbach, M.; Molinari, F.; Beek, E.J.R.V.; Biederer, J. MRI of the lung (1/3): Methods. *Insights Imaging* **2012**, *3*, 345–353. [CrossRef]
- 60. Kumar, S.; Liney, G.; Rai, R.; Holloway, L.; Moses, D.; Vinod, S.K. Magnetic resonance imaging in lung: A review of its potential for radiotherapy. *Br. J. Radiol.* 2016, *89*, 20150431. [CrossRef]
- Hatabu, H.; Ohno, Y.; Gefter, W.B.; Parraga, G.; Madore, B.; Lee, K.S.; Altes, T.A.; Lynch, D.A.; Mayo, J.R.; Seo, J.B.; et al. Expanding Applications of Pulmonary MRI in the Clinical Evaluation of Lung Disorders: Fleischner Society Position Paper. *Radiology* 2020, 297, 286–301. [CrossRef]
- 62. Jiang, W.; Ong, F.; Johnson, K.M.; Nagle, S.K.; Hope, T.A.; Lustig, M.; Larson, P.E. Motion robust high resolution 3D free-breathing pulmonary MRI using dynamic 3D image self-navigator. *Magn. Reson. Med.* **2017**, *79*, 2954–2967. [CrossRef]
- Meier-Schroers, M.; Homsi, R.; Gieseke, J.; Schild, H.H.; Thomas, D. Lung cancer screening with MRI: Evaluation of MRI for lung cancer screening by comparison of LDCT- and MRI-derived Lung-RADS categories in the first two screening rounds. *Eur. Radiol.* 2018, 29, 898–905. [CrossRef]
- 64. Li, B.; Li, Q.; Chen, C.; Guan, Y.; Liu, S. A Systematic Review and Meta-analysis of the Accuracy of Diffusion-weighted MRI in the Detection of Malignant Pulmonary Nodules and Masses. *Acad. Radiol.* **2014**, *21*, 21–29. [CrossRef] [PubMed]
- 65. Luna, A.; Sánchez-Gonzalez, J.; Caro, P. Diffusion-Weighted Imaging of the Chest. *Magn. Reson. Imaging Clin. N. Am.* 2011, 19, 69–94. [CrossRef] [PubMed]
- Razek, A.A.K.A.; Soliman, N.; Elashery, R. Apparent diffusion coefficient values of mediastinal masses in children. *Eur. J. Radiol.* 2012, *81*, 1311–1314. [CrossRef]
- Usuda, K.; Iwai, S.; Yamagata, A.; Sekimura, A.; Motono, N.; Matoba, M.; Doai, M.; Yamada, S.; Ueda, Y.; Hirata, K.; et al. Relationships and Qualitative Evaluation between Diffusion-Weighted Imaging and Pathologic Findings of Resected Lung Cancers. *Cancers* 2020, *12*, 1194. [CrossRef]
- Cheran, S.K.; Nielsen, N.D.; Patz, E.F. False-Negative Findings for Primary Lung Tumors on FDG Positron Emission Tomography:Staging and Prognostic Implications. *Am. J. Roentgenol.* 2004, 182, 1129–1132. [CrossRef] [PubMed]
- Usuda, K.; Sagawa, M.; Motono, N.; Ueno, M.; Tanaka, M.; Machida, Y.; Maeda, S.; Matoba, M.; Kuginuki, Y.; Taniguchi, M.; et al. Diagnostic Performance of Diffusion Weighted Imaging of Malignant and Benign Pulmonary Nodules and Masses: Comparison with Positron Emission Tomography. *Asian Pac. J. Cancer Prev.* 2014, 15, 4629–4635. [CrossRef]
- Mori, T.; Nomori, H.; Ikeda, K.; Kawanaka, K.; Shiraishi, S.; Katahira, K.; Yamashita, Y. Diffusion-Weighted Magnetic Resonance Imaging for Diagnosing Malignant Pulmonary Nodules/Masses: Comparison with Positron Emission Tomography. J. Thorac. Oncol. 2008, 3, 358–364. [CrossRef] [PubMed]
- 71. Zhang, J.; Cui, L.B.; Tang, X.; Ren, X.L.; Shi, J.R.; Yang, H.N.; Zhang, Y.; Li, Z.K.; Wu, C.G.; Jian, W.; et al. DW MRI at 3.0 T versus FDG PET/CT for detection of malignant pulmonary tumors. *Int. J. Cancer* **2013**, *134*, 606–611. [CrossRef]
- 72. Nomori, H.; Cong, Y.; Sugimura, H.; Kato, Y. Comparing diffusion-weighted imaging and positron emission tomography for pulmonary nodules measuring from 1 to 3 cm in size. *Surg. Today* **2015**, *45*, 1535–1541. [CrossRef]
- 73. Wu, L.M.; Xu, J.R.; Hua, J.; Gu, H.Y.; Chen, J.; Haacke, E.; Hu, J. Can diffusion-weighted imaging be used as a reliable sequence in the detection of malignant pulmonary nodules and masses? *Magn. Reson. Imaging* **2013**, *31*, 235–246. [CrossRef]
- Peerlings, J.; Troost, E.G.C.; Nelemans, P.J.; Cobben, D.C.P.; de Boer, J.C.J.; Hoffmann, A.L.; Beets-Tan, R.G.H. The Diagnostic Value of MR Imaging in Determining the Lymph Node Status of Patients with Non–Small Cell Lung Cancer: A Meta-Analysis. *Radiology* 2016, 281, 86–98. [CrossRef] [PubMed]
- Shen, G.; Hu, S.; Deng, H.; Kuang, A. Performance of DWI in the Nodal Characterization and Assessment of Lung Cancer: A Meta-Analysis. Am. J. Roentgenol. 2016, 206, 283–290. [CrossRef]
- Guan, H.X.; Pan, Y.Y.; Wang, Y.J.; Tang, D.Z.; Zhou, S.C.; Xia, L.M. Comparison of Various Parameters of DWI in Distinguishing Solitary Pulmonary Nodules. *Curr. Med Sci.* 2018, 38, 920–924. [CrossRef]
- 77. Colletti, P.M. Reverse Phase Encoding–corrected DWI Improves MRI for PET/MRI of Lung Cancer. *Radiology* **2020**, 295, 701–702. [CrossRef] [PubMed]
- 78. Wang, L.L.; Lin, J.; Liu, K.; Chen, C.Z.; Liu, H.; Lv, P.; Fu, C.X.; Zeng, M.S. Intravoxel incoherent motion diffusion-weighted MR imaging in differentiation of lung cancer from obstructive lung consolidation: Comparison and correlation with pharmacokinetic analysis from dynamic contrast-enhanced MR imaging. *Eur. Radiol.* 2014, 24, 1914–1922. [CrossRef]
- 79. Xiao, Z.; Zhong, Y.; Tang, Z.; Qiang, J.; Qian, W.; Wang, R.; Wang, J.; Wu, L.; Tang, W.; Zhang, Z. Standard diffusion-weighted, diffusion kurtosis and intravoxel incoherent motion MR imaging of sinonasal malignancies: Correlations with Ki-67 proliferation status. *Eur. Radiol.* **2018**, *28*, 2923–2933. [CrossRef] [PubMed]

- Wen, S.; Zhou, W.; Li, C.M.; Hu, J.; Hu, X.M.; Chen, P.; Shao, G.L.; Guo, W.H. Ki-67 as a prognostic marker in early-stage non-small cell lung cancer in Asian patients: A meta-analysis of published studies involving 32 studies. *BMC Cancer* 2015, *15*, 520. [CrossRef]
- Wei, D.M.; Chen, W.J.; Meng, R.M.; Zhao, N.; Zhang, X.Y.; Liao, D.Y.; Chen, G. Augmented expression of Ki-67 is correlated with clinicopathological characteristics and prognosis for lung cancer patients: An up-dated systematic review and meta-analysis with 108 studies and 14,732 patients. *Respir. Res.* 2018, 19, 150. [CrossRef]
- Besusparis, J.; Plancoulaine, B.; Rasmusson, A.; Augulis, R.; Green, A.R.; Ellis, I.O.; Laurinaviciene, A.; Herlin, P.; Laurinavicius, A. Impact of tissue sampling on accuracy of Ki67 immunohistochemistry evaluation in breast cancer. *Diagn. Pathol.* 2016, 11, 82. [CrossRef] [PubMed]
- 83. Zheng, Y.; Huang, W.; Zhang, X.; Lu, C.; Fu, C.; Li, S.; Lin, G. A Noninvasive Assessment of Tumor Proliferation in Lung cancer Patients using Intravoxel Incoherent Motion Magnetic Resonance Imaging. *J. Cancer* **2021**, *12*, 190–197. [CrossRef]
- Kono, R.; Fujimoto, K.; Terasaki, H.; Müller, N.L.; Kato, S.; Sadohara, J.; Hayabuchi, N.; Takamori, S. Dynamic MRI of Solitary Pulmonary Nodules: Comparison of Enhancement Patterns of Malignant and Benign Small Peripheral Lung Lesions. *Am. J. Roentgenol.* 2007, 188, 26–36. [CrossRef]
- Schaefer, J.F.; Vollmar, J.; Schick, F.; Vonthein, R.; Seemann, M.D.; Aebert, H.; Dierkesmann, R.; Friedel, G.; Claussen, C.D. Solitary Pulmonary Nodules: Dynamic Contrast-enhanced MR Imaging—Perfusion Differences in Malignant and Benign Lesions. *Radiology* 2004, 232, 544–553. [CrossRef] [PubMed]
- Ohno, Y.; Nishio, M.; Koyama, H.; Seki, S.; Tsubakimoto, M.; Fujisawa, Y.; Yoshikawa, T.; Matsumoto, S.; Sugimura, K. Solitary Pulmonary Nodules: Comparison of Dynamic First-Pass Contrast-enhanced Perfusion Area-Detector CT, Dynamic First-Pass Contrast-enhanced MR Imaging, and FDG PET/CT. *Radiology* 2015, 274, 563–575. [CrossRef]
- 87. Koyama, H.; Ohno, Y.; Seki, S.; Nishio, M.; Yoshikawa, T.; Matsumoto, S.; Sugimura, K. Magnetic Resonance Imaging for Lung Cancer. J. Thorac. Imaging 2013, 28, 138–150. [CrossRef]
- Zou, Y.; Zhang, M.; Wang, Q.; Shang, D.; Wang, L.; Yu, G. Quantitative Investigation of Solitary Pulmonary Nodules: Dynamic Contrast-Enhanced MRI and Histopathologic Analysis. *Am. J. Roentgenol.* 2008, 191, 252–259. [CrossRef]
- Roos, J.E.; McAdams, H.P.; Kaushik, S.S.; Driehuys, B. Hyperpolarized Gas MR Imaging. Magn. Reson. Imaging Clin. N. Am. 2015, 23, 217–229. [CrossRef] [PubMed]
- McMahon, C.J.; Dodd, J.D.; Hill, C.; Woodhouse, N.; Wild, J.M.; Fichele, S.; Gallagher, C.G.; Skehan, S.J.; van Beek, E.J.R.; Masterson, J.B. Hyperpolarized 3helium magnetic resonance ventilation imaging of the lung in cystic fibrosis: Comparison with high resolution CT and spirometry. *Eur. Radiol.* 2006, 16, 2483–2490. [CrossRef]
- 91. Mathew, L.; Wheatley, A.; Castillo, R.; Castillo, E.; Rodrigues, G.; Guerrero, T.; Parraga, G. Hyperpolarized 3He Magnetic Resonance Imaging. *Acad. Radiol.* **2012**, *19*, 1546–1553. [CrossRef] [PubMed]
- Ireland, R.H.; Bragg, C.M.; McJury, M.; Woodhouse, N.; Fichele, S.; van Beek, E.J.; Wild, J.M.; Hatton, M.Q. Feasibility of Image Registration and Intensity-Modulated Radiotherapy Planning With Hyperpolarized Helium-3 Magnetic Resonance Imaging for Non–Small-Cell Lung Cancer. Int. J. Radiat. Oncol. Biol. Phys. 2007, 68, 273–281. [CrossRef]
- 93. Li, B.; Li, Q.; Nie, W.; Liu, S. Diagnostic value of whole-body diffusion-weighted magnetic resonance imaging for detection of primary and metastatic malignancies: A meta-analysis. *Eur. J. Radiol.* **2014**, *83*, 338–344. [CrossRef]
- Medeiros, T.M.; Altmayer, S.; Watte, G.; Zanon, M.; Dias, A.B.; Concatto, N.H.; Paes, J.H.; Mattiello, R.; de Souza Santos, F.; Mohammed, T.L.; et al. 18F-FDG PET/CT and whole-body MRI diagnostic performance in M staging for non–small cell lung cancer: A systematic review and meta-analysis. *Eur. Radiol.* 2020, *30*, 3641–3649. [CrossRef] [PubMed]
- Lee, D.H.; Lee, J.M. Whole-body PET/MRI for colorectal cancer staging: Is it the way forward? J. Magn. Reson. Imaging 2016, 45, 21–35. [CrossRef] [PubMed]
- Dahlsgaard-Wallenius, S.E.; Hildebrandt, M.G.; Johansen, A.; Vilstrup, M.H.; Petersen, H.; Gerke, O.; Høilund-Carlsen, P.F.; Morsing, A.; Andersen, T.L. Hybrid PET/MRI in non-small cell lung cancer (NSCLC) and lung nodules—A literature review. *Eur. J. Nucl. Med. Mol. Imaging* 2020, 48, 584–591. [CrossRef]
- Brierley, J.D.; Gospodarowicz, M.K.; Wittekind, C. TNM Classification of Malignant Tumours; John Wiley & Sons: Hoboken, NJ, USA, 2017.
- 98. Feng, S.H.; Yang, S.T. The new 8th TNM staging system of lung cancer and its potential imaging interpretation pitfalls and limitations with CT image demonstrations. *Diagn. Interv. Radiol.* **2019**, *25*, 270–279. [CrossRef] [PubMed]
- 99. Travis, W.D.; Asamura, H.; Bankier, A.A.; Beasley, M.B.; Detterbeck, F.; Flieder, D.B.; Goo, J.M.; MacMahon, H.; Naidich, D.; Nicholson, A.G.; et al. The IASLC Lung Cancer Staging Project: Proposals for Coding T Categories for Subsolid Nodules and Assessment of Tumor Size in Part-Solid Tumors in the Forthcoming Eighth Edition of the TNM Classification of Lung Cancer. J. Thorac. Oncol. 2016, 11, 1204–1223. [CrossRef]
- 100. Kajiwara, N.; Akata, S.; Uchida, O.; Usuda, J.; Ohira, T.; Kawate, N.; Ikeda, N. Cine MRI enables better therapeutic planning than CT in cases of possible lung cancer chest wall invasion. *Lung Cancer* **2010**, *69*, 203–208. [CrossRef] [PubMed]
- Rami-Porta, R.; Call, S.; Dooms, C.; Obiols, C.; Sánchez, M.; Travis, W.D.; Vollmer, I. Lung cancer staging: A concise update. *Eur. Respir. J.* 2018, 51, 1800190. [CrossRef]
- 102. Goldstraw, P.; Chansky, K.; Crowley, J.; Rami-Porta, R.; Asamura, H.; Eberhardt, W.E.; Nicholson, A.G.; Groome, P.; Mitchell, A.; Bolejack, V.; et al. The IASLC Lung Cancer Staging Project: Proposals for Revision of the TNM Stage Groupings in the Forthcoming (Eighth) Edition of the TNM Classification for Lung Cancer. J. Thorac. Oncol. 2016, 11, 39–51. [CrossRef]

- 103. Mb, A. American Cancer Society. In *AJCC Cancer Staging Manual*, 8th ed.; American Joint Committee on Cancer: Chicago, IL, USA, 2017.
- 104. Asamura, H.; Chansky, K.; Crowley, J.; Goldstraw, P.; Rusch, V.W.; Vansteenkiste, J.F.; Watanabe, H.; Wu, Y.L.; Zielinski, M.; Ball, D.; et al. The International Association for the Study of Lung Cancer Lung Cancer Staging Project. *J. Thorac. Oncol.* 2015, 10, 1675–1684. [CrossRef]
- Zhao, L.; He, Z.Y.; Zhong, X.N.; Cui, M.L. 18FDG-PET/CT for detection of mediastinal nodal metastasis in non-small cell lung cancer: A meta-analysis. *Surg. Oncol.* 2012, 21, 230–236. [CrossRef]
- 106. Wu, L.M.; Xu, J.R.; Gu, H.Y.; Hua, J.; Chen, J.; Zhang, W.; Haacke, E.M.; Hu, J. Preoperative mediastinal and hilar nodal staging with diffusion-weighted magnetic resonance imaging and fluorodeoxyglucose positron emission tomography/computed tomography in patients with non-small-cell lung cancer: Which is better? J. Surg. Res. 2012, 178, 304–314. [CrossRef] [PubMed]
- 107. Pauls, S.; Schmidt, S.A.; Juchems, M.S.; Klass, O.; Luster, M.; Reske, S.N.; Brambs, H.J.; Feuerlein, S. Diffusion-weighted MR imaging in comparison to integrated [18F]-FDG PET/CT for N-staging in patients with lung cancer. *Eur. J. Radiol.* 2012, *81*, 178–182. [CrossRef]
- 108. Milovanovic, I.; Stjepanovic, M.; Mitrovic, D. Distribution patterns of the metastases of the lung carcinoma in relation to histological type of the primary tumor: An autopsy study. *Ann. Thorac. Med.* **2017**, *12*, 191. [CrossRef] [PubMed]
- Millon, D.; Byl, D.; Collard, P.; Cambier, S.E.; Maanen, A.G.V.; Vlassenbroek, A.; Coche, E.E. Could new reconstruction CT techniques challenge MRI for the detection of brain metastases in the context of initial lung cancer staging? *Eur. Radiol.* 2017, 28, 770–779. [CrossRef] [PubMed]
- Kara, P.O.; Kara, T.; Gedik, G.K.; Kara, F.; Sahin, O.; Gunay, E.C.; Sari, O. The role of fluorodeoxyglucose-positron emission tomography/computed tomography in differentiating between benign and malignant adrenal lesions. *Nucl. Med. Commun.* 2011, 32, 106–112. [CrossRef] [PubMed]
- Nishino, M.; Hatabu, H.; Johnson, B.E.; McLoud, T.C. State of the art: Response assessment in lung cancer in the era of genomic medicine. *Radiology* 2014, 271, 6–27. [CrossRef] [PubMed]
- 112. Wahl, R.L.; Jacene, H.; Kasamon, Y.; Lodge, M.A. From RECIST to PERCIST: Evolving Considerations for PET Response Criteria in Solid Tumors. J. Nucl. Med. 2009, 50, 1225–150S. [CrossRef]
- 113. Usuda, K.; Iwai, S.; Funasaki, A.; Sekimura, A.; Motono, N.; Matoba, M.; Doai, M.; Yamada, S.; Ueda, Y.; Uramoto, H. Diffusionweighted magnetic resonance imaging is useful for the response evaluation of chemotherapy and/or radiotherapy to recurrent lesions of lung cancer. *Transl. Oncol.* **2019**, *12*, 699–704. [CrossRef]
- 114. William, W.N.; Pataer, A.; Kalhor, N.; Correa, A.M.; Rice, D.C.; Wistuba, I.I.; Heymach, J.; Lee, J.J.; Kim, E.S.; Munden, R.; et al. Computed Tomography RECIST Assessment of Histopathologic Response and Prediction of Survival in Patients with Resectable Non–Small-Cell Lung Cancer after Neoadjuvant Chemotherapy. J. Thorac. Oncol. 2013, 8, 222–228. [CrossRef] [PubMed]
- 115. Dunlap, N.E.; Yang, W.; McIntosh, A.; Sheng, K.; Benedict, S.H.; Read, P.W.; Larner, J.M. Computed Tomography-Based Anatomic Assessment Overestimates Local Tumor Recurrence in Patients with Mass-like Consolidation After Stereotactic Body Radiotherapy for Early-Stage Non-Small Cell Lung Cancer. Int. J. Radiat. Oncol. Biol. Phys. 2012, 84, 1071–1077. [CrossRef]
- 116. Wang, J.; Wong, K.K.; Piert, M.; Stanton, P.; Frey, K.A.; Kong, F.M. Metabolic response assessment with 18F-FDG PET/CT: Inter-method comparison and prognostic significance for patients with non-small cell lung cancer. *J. Radiat. Oncol.* 2015, 4, 249–256. [CrossRef]
- 117. Na, F.; Wang, J.; Li, C.; Deng, L.; Xue, J.; Lu, Y. Primary Tumor Standardized Uptake Value Measured on F18-Fluorodeoxyglucose Positron Emission Tomography Is of Prediction Value for Survival and Local Control in Non–Small-Cell Lung Cancer Receiving Radiotherapy: Meta-Analysis. J. Thorac. Oncol. 2014, 9, 834–842. [CrossRef] [PubMed]
- Ohno, Y.; Koyama, H.; Yoshikawa, T.; Matsumoto, K.; Aoyama, N.; Onishi, Y.; Sugimura, K. Diffusion-Weighted MRI Versus 18F-FDG PET/CT: Performance as Predictors of Tumor Treatment Response and Patient Survival in Patients With Non–Small Cell Lung Cancer Receiving Chemoradiotherapy. Am. J. Roentgenol. 2012, 198, 75–82. [CrossRef]
- Yabuuchi, H.; Hatakenaka, M.; Takayama, K.; Matsuo, Y.; Sunami, S.; Kamitani, T.; Jinnouchi, M.; Sakai, S.; Nakanishi, Y.; Honda, H. Non–Small Cell Lung Cancer: Detection of Early Response to Chemotherapy by Using Contrast-enhanced Dynamic and Diffusion-weighted MR Imaging. *Radiology* 2011, 261, 598–604. [CrossRef] [PubMed]
- Nunes, T.F.; Szejnfeld, D.; Szejnfeld, J.; Kater, C.E.; Faintuch, S.; Castro, C.H.M.; Goldman, S.M. Assessment of Early Treatment Response With DWI After CT-Guided Radiofrequency Ablation of Functioning Adrenal Adenomas. *Am. J. Roentgenol.* 2016, 207, 804–810. [CrossRef] [PubMed]
- Park, H.; Sholl, L.M.; Hatabu, H.; Awad, M.M.; Nishino, M. Imaging of Precision Therapy for Lung Cancer: Current State of the Art. *Radiology* 2019, 293, 15–29. [CrossRef] [PubMed]
- Gazdar, A.F. Personalized Medicine and Inhibition of EGFR Signaling in Lung Cancer. N. Engl. J. Med. 2009, 361, 1018–1020. [CrossRef]
- Nishino, M.; Jackman, D.M.; Hatabu, H.; Jänne, P.A.; Johnson, B.E.; den Abbeele, A.D.V. Imaging of Lung Cancer in the Era of Molecular Medicine. Acad. Radiol. 2011, 18, 424–436. [CrossRef]
- 124. Keedy, V.L.; Temin, S.; Somerfield, M.R.; Beasley, M.B.; Johnson, D.H.; McShane, L.M.; Milton, D.T.; Strawn, J.R.; Wakelee, H.A.; Giaccone, G. American Society of Clinical Oncology Provisional Clinical Opinion: Epidermal Growth Factor Receptor (EGFR) Mutation Testing for Patients With Advanced Non–Small-Cell Lung Cancer Considering First-Line EGFR Tyrosine Kinase Inhibitor Therapy. J. Clin. Oncol. 2011, 29, 2121–2127. [CrossRef]

- 125. Kwak, E.L.; Bang, Y.J.; Camidge, D.R.; Shaw, A.T.; Solomon, B.; Maki, R.G.; Ou, S.H.I.; Dezube, B.J.; Jänne, P.A.; Costa, D.B.; et al. Anaplastic Lymphoma Kinase Inhibition in Non–Small-Cell Lung Cancer. *N. Engl. J. Med.* **2010**, *363*, 1693–1703. [CrossRef]
- 126. Ettinger, D.S.; Akerley, W.; Borghaei, H.; Chang, A.C.; Cheney, R.T.; Chirieac, L.R.; D'Amico, T.A.; Demmy, T.L.; Ganti, A.K.P.; Govindan, R.; et al. Non–Small Cell Lung Cancer. *J. Natl. Compr. Cancer Netw.* **2012**, *10*, 1236–1271. [CrossRef] [PubMed]